Author's response to reviews

Title: Efficiency of two-phase methods with focus on a planned population-based case-control study on air pollution and stroke

Authors:

Anna Oudin (anna.oudin@med.lu.se)
Jonas Bjork (jonas.bjork@skane.se)
Ulf Stromberg (ulf.stromberg@med.lu.se)

Version: 7 Date: 26 October 2007

Author's response to reviews:

Dear Editors,

I've made formatting changes according to your suggestions.
I've changed the tables so that there is now no merged cells that crosses rows.
I've also changed the description of the additional files, and changed the text when referring to the additional files.
I hope I was successful in changing the manuscript according to your wishes.

Sincerely

Anna Oudin